Author's response to reviews

Title: Why do women present late for antenatal care? A qualitative interview study

Authors:

Rosalind Haddrill (r.haddrill@sheffield.ac.uk)
Georgina L Jones (g.l.jones@sheffield.ac.uk)
Caroline A Mitchell (c.mitchell@sheffield.ac.uk)
Dilly OC Anumba (d.o.c.anumba@sheffield.ac.uk)

Version: 3 Date: 2 June 2014

Author's response to reviews: see over
Dear Editor,

Re: Why do women present late for antenatal care? A qualitative interview study

We would be very grateful if you would consider our revised manuscript for publication in BMC Pregnancy and Childbirth. We have attached the manuscript which has been formatted in accordance with the 'BMC-series medical journals - authors' checklist for manuscript formatting' guidance. We have also detailed our responses to the editor’s comments, with the appropriate page numbers, below.

1. Title page
The title page has been amended according to the BMC Pregnancy and Childbirth guidance provided; extra information has been removed, as requested. An authors’ information section has been created later in the document, as suggested (p39).

2. List of abbreviations
The abbreviations used in the text have been defined at first use, and a list of abbreviations provided preceding the competing interests and authors' contributions, as suggested (p37).

We hope your editorial team and reviewers will give our work your full consideration.

Yours sincerely,

Ms Rosalind Haddrill
Dr Georgina Jones
Dr Caroline Mitchell
Professor Dilly Anumba